Author's response to reviews

Title: Neutral sphingomyelinase 2 modulates cytotoxic effects of protopanaxadiol on different human cancer cells

Authors:

Bonggoo Park (parkbg35@gmail.com)
Yong-Moon Lee (ymleefn@chungbuk.ac.kr)
Jae-Sung Kim (jae.kim@surgery.ufl.edu)
Youl Her (marketing@btgin.com)
Ju Hee Kang (applekjh0503@hanmail.net)
Seung-Hyun Oh (eyeball@gachon.ac.kr)
Hwan-Mook Kim (hwanmook@gachon.ac.kr)

Version: 2 Date: 23 January 2013

Author's response to reviews: see over
Cover Letter

Dear Sir or Madam,

Thank you for suggesting changes in “Competing Interest” section to declare that the authors have no competing interests. We have made these changes in our revised manuscript. Please reconsider our manuscript for publication in your journal.

Enclosed is a manuscript titled “Neutral sphingomyelinase 2 modulates cytotoxic effects of protopanaxadiol on different human cancer cells” for consideration in your journal. This manuscript presents a novel finding among the molecular mechanisms by which the ginsenoside exerts cytotoxicity against various cancer cells. Protopanaxadiol (PPD) had cytotoxic effects against two distinct cancer cells through common mechanism(s) including the activation of neutral sphingomyelinase 2 and thereby production of intracellular ceramides. Thus, this novel mechanism could be employed as potential drug targets in cancer therapeutics.

All the findings in this manuscript are original and have not been presented in any official meeting till now. This manuscript has not been published nor accepted for publication in any journal. And this research is not under consideration for publication in your or any other journal elsewhere at the moment.

We have carefully followed your author guidelines in preparation of our manuscript for formatting and style, and we look forward to hearing your comments on our work.

Conflict of interest
The authors declare no conflict of interest.

Yours truly,

Hwan Mook Kim, PhD
Gachon University
College of Pharmacy